Author’s response to reviews

Title: Health related quality of life after oesophagectomy for cancer in the elderly

Authors:

Francesco Cavallin (cescocava@libero.it)
Eleonora Pinto (eleonora.pinto@ioveneto.it)
Luca M Saadeh (luca.saadeh@gmail.com)
Rita Alfieri (ritalalfieri@ioveneto.it)
Matteo Cagol (matteo.cagol@unipd.it)
Carlo Castoro (carlo.castoro@unipd.it)
Marco Scarpa (marcoscarpa73@yahoo.it)

Version: 2 Date: 9 January 2015

Author’s response to reviews:

1. Line Numbering.

The revised manuscript now includes line and page numbers. Page breaks (MS Word format) have been removed as requested.

2. Acknowledgements.

The sources of funding have been included in the Acknowledgements Section. The funding bodies had no role in the design of the study, in the collection, analysis, and interpretation of data, in the writing of the manuscript and in the decision to submit the manuscript for publication.